Author's response to reviews

Title: Bioelectrical impedance phase angle in clinical practice: implications for prognosis in stage IIIB and IV non-small cell lung cancer

Authors:

Digant Gupta (gupta_digant@yahoo.com)
Carolyn A Lammersfeld (Carolyn.lammersfeld@ctca-hope.com)
Pankaj G Vashi (pgvashi@aol.com)
Jessica King (Jessica.king@ctca-hope.com)
Sadie L Dahlk (sadie.dahlk@ctca-hope.com)
James F Grutsch (jfgrutsch@yahoo.com)
Christopher G Lis (Christopher.lis@ctca-hope.com)

Version: 2 Date: 30 July 2008

Author's response to reviews:

1) Scope: Given that your new submission relates to lung cancer and is closely related to your BMC Cancer manuscript, we feel it would be better suited to BMC Cancer rather than BMC Public Health. If you would be happy to have your manuscript transferred to BMC Cancer for consideration then please let us know as soon as possible and we will handle the transfer on your behalf. PLEASE FEEL FREE TO TRANSFER THE MANUSCRIPT TO BMC CANCER.

2) Overlap: We would like you to avoid duplicating the methods, and other parts the new manuscript, and instead cite the methods that have already been described in your previous manuscript. Please also clarify in the lung cancer manuscript, whether the data was collected at the same time as that of the breast cancer manuscript. WE HAVE CITED OUR BREAST CANCER PAPER IN THE METHODS AND DISCUSSION SECTIONS OF THE REVISED PAPER TO AVOID DUPLICATION OF TEXT.

3) Ethical approval: Please clarify the ethical approval for your research. Experimental research that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/e/policy/b3.htm), and any experimental research on animals must follow internationally recognized guidelines. A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate. STATEMENT ON ETHICAL APPROVAL ADDED UNDER THE METHODS SECTION.

4) Authors’ contributions - Please include an Authors’ contributions section before the Acknowledgements and Reference list. AUTHORS’ CONTRIBUTION SECTION ADDED.